Author’s response to reviews

Title: Surgical offloading procedures for diabetic foot ulcers compared to best non-surgical treatment: A study protocol for a randomized controlled trial

Authors:

Aharon Finestone (asff@inter.net.il)
Eran Tamir (eranetta0@gmail.com)
Guy Ron Tzvi (guy_ron_tzvi@hotmail.com)
Itay Wiser (wiser125@gmail.com)
Gabriel Agar (gabyagar@gmail.com)

Version: 2 Date: 02 Feb 2018

Author’s response to reviews:

Attached in personal cover letter.